Author's response to reviews

Title: Tear of Peroneus longus in long distance runners due to enlarged Peroneal tubercle.

Authors:

Ezequiel Palmanovich (ezepalm@gmail.com)
Lior Laver (laver17@gmail.com)
Yaron S Brin (yaronbrin@gmail.com)
Iftach Hetsrony (hetsrony@netvision.net.il)
Meir Nyska Prof (nyska@internet-zahav.net)

Version: 4 Date: 30 July 2012

Author's response to reviews: see over
Dear SMARTT TEAM

Thank you for the opportunity to submit our revised manuscript to SMART. In response to the reviewers’ comments, which are attached below, the paper was reviewed and revised by an English language medical editor. It was also formatted according to the journal instructions.

Comments:

Figure 1 is not clear. What is the tendon and what is synovitis? Using arrows and circles, give more clear explanation with the relation of bony structures and tendon and synovitis.

In Figure 2, the authors should indicate some relationship between injured tendon with synovitis and bony prominence.

Figure 3 and 4 are same figures. Figure legends of figures 3 and 4 do not make sense to the readers. Give more clear explanation with the relation of bony structures to tendon tear and enlarged peroneal tubercle.

Figure 7 is also out of focus. Please change the picture or add the explanation on the picture.

As the reviewer request all the item was corrected. Also new photo was introduced and the not clear were removed.

Please let me know if there are any other questions or concerns. Thank you.

Corresponding author

Dr Palmanovich Ezequiel
Orthopedics Department
Meir Medical Center
Tchernichovsky st. 59
Kfar-Saba - 44281
Israel
Cell phone: 052-6439595
Email: ezequiel.palmanovich@clalit.org.il
ezepalm@gmail.com